Author's response to reviews

Title: Evolving the theory and praxis of knowledge translation through social interaction: A social phenomenological study

Authors:

Carol L McWilliam Dr. (cmcwill@uwo.ca)
Anita Kothari Dr. (akothari@uwo.ca)
Catherine Ward-Griffin Dr. (cwg@uwo.ca)
Dorothy Forbes Dr. (dforbes6@uwo.ca)
Beverley Leipert Dr. (bleipert@uwo.ca)
SW-CCAC Home Care Collaboration the (Sandra.Coleman@sw.ccac-ont.ca)

Version: 4 Date: 22 April 2009

Author's response to reviews:

Dear. Dr. Eccles: Please note that we have accepted all recommended changes, and made revisions in answer to the two questions. We have also made two additional edits previously overlooked, specifically, on p. 14 of the tracked version, under "Sample", line 1, we've inserted a missing word, namely "constructed", following "geographically-". In the first quotation on p. 17 of the tracked version, line 5, we've revised the wording of the boldfaced interpretation to create parallelism with other interpretive notations, so that "...a facilitator as an organization" has been changed to read "... a macro/organizational facilitator to KT". We trust that it is not too late to do so. If this creates problems, please disregard.- Sincerely, Carol McWilliam, Corresponding Author